CS833: Advanced Topics in Operating Systems

Fall, 1999

Instructor: Howard J. Hamilton, CW308.21, #585-4079, hamilton@cs.uregina.ca Lectures: Monday, Wednesday, Friday 9:30-10:20 am, UC286 Office Hours: MWF 10:20 – 11:20 & Thursday, 2:30 – 3:00 pm At other times, please send email.

Reference Books:

1. Stevens, W.R., UNIX Network Programming, Prentice-Hall, 1990. 2. A.S. Tanenbaum, Distributed Operating Systems, Prentice-Hall, 1995. 3. J. S. Gray, Interprocess Communications in UNIX: The Nooks and Crannies.

Prentice-Hall, 1997. (Alternate to Stevens.)

4. A. Silberschatz and P. Galvin, Operating Systems Concepts, 4th edition, Addison Wesley, 1994.

Grading

50% — grade from CS430 without project 10% — project proposal, to be submitted Oct. 22, 1999 40% — final project report (oral and written), oral to be given December 7, 1999; written to be submitted December 13, 1999